Author's response to reviews

Title: High resolution human leukocyte antigen (HLA) class I and class II allele typing in Mexican mestizo women with sporadic breast cancer: case-control study.

Authors:

David Cantu de Leon (dcantude@yahoo.com)
Delia Perez Montiel (madeliapmg@hotmail.com)
Veronica Villavicencio (verovilla67@hotmail.com)
Alejandro García Carranca (carranca@biomedicas.unam.mx)
Alejandro Mohar Betancourt (amohar@incan.edu.mx)
Victor Acuña-Alonzo (acunaav@gmail.com)
Alberto López-Tello (lopeztelloa@gmeil.com)
Gilberto Vargas-Alarcón (gvargas63@yahoo.com)
Rodrigo Barquera (jrockdrigo@excite.com)
Neng Yu (yun@usa.redcross.org)
Edmond J Yunis (Edmond_Yunis@dfci.harvard.edu)
Julio Granados (julgrate@yahoo.com)

Version: 4 Date: 1 February 2009

Author's response to reviews: see over

Rikki Graham, PhD  
Senior Assistant Editor  
BMC-series journals

MS: 1899294259226733: High resolution human leukocyte antigen (HLA) class I and class II allele typing in Mexican mestizo women with sporadic breast cancer: case-control study.

Dear Doctor Graham:

According to the request about formatting requirements we did the next changes:
1.- We check the list of authors and is correct now.

2.- We check the abstract and now match exactly.

3.- We re named the section “Methods”.

4.- Heading Competing interests was done.

Thank you

Regards,

David Cantú de León  
Department of Gynecologic Oncology.  
Instutito Nacional de Cancerologia, Mexico.